Author's response to reviews

Title: miR-320b suppresses cell proliferation by targeting c-Myc in human colorectal cancer cells

Authors:

Hantao Wang (hantaowang@126.com)
Fuao Cao (fuaocao2003@163.com)
Xu Li (xuli_ch@163.com)
Hua Miao (13567330917@139.com)
Jifu E (jefershon@sohu.com)
Junjie Xing (xjj2118@163.com)
Chuan-gang Fu (fugang416@126.com)

Version: 5
Date: 20 May 2015

Author's response to reviews: see over
Dear Ryan Relox:

We appreciate your time reviewing our manuscript MS: 2079497320154697 - miR-320b suppresses cell proliferation by targeting c-Myc in human colorectal cancer cells; and we have made revisions in accordance with your suggestions as follows:

1. The manuscript has been carefully reformatted to your journal style, according to the link you provided in the previous email.

2. The language has been edited by a native-English speaker with medical science expertise from a professional language editing service.

Yours sincerely,

Chuan-gang Fu, M.D.
Professor and Director, Department of Colorectal Surgery, Changhai Hospital, Shanghai 200433, China
Email: fugang416@126.com